Author's response to reviews

Title: Advanced biliary tract carcinomas: a retrospective multicenter analysis of first and second-line chemotherapy

Authors:

Frederic Fiteni (fredericfiteni@gmail.com)
Marine Jary (jary.marine@yahoo.fr)
Franck Monnien (franck.monnien@gmail.com)
Thierry Nguyen (tnguyen@chu-besancon.fr)
Eric Beohou (eric.beohou@gmail.com)
Martin Demarchi (mdemarchi@chu-besancon.fr)
Erion Dobi (erion.dr@yahoo.com)
Francine Fein (ffein@chu-besancon.fr)
Denis Cleau (d.cleau@chi70.fr)
Serge Fratté (sfratte@chbm.fr)
Virginie Nerich (v1nerich@chu-besancon.fr)
Franck Bonnetain (franck.bonnetain@univ-fcomte.fr)
Xavier Pivot (xavier.pivot@univ-fcomte.fr)
Christophe Borg (christophe.borg@efs.sante.fr)
Stefano Kim (stefanokim@gmail.com)

Version: 3
Date: 4 August 2014

Author's response to reviews: see over
Dear Editorial Committee,

We are very pleased to submit the final version of our manuscript written by Fiteni F. et al. entitled “Advanced biliary tract carcinomas: a retrospective multicenter analysis of first and second-line chemotherapy”.

All editorial comments were taken into account in this revised manuscript.

Please find below a point-by-point response to your comments.

1) Please copy edit your manuscript,

2) Please include the full name of the ethics committee that approved your study in the methods section of the manuscript,

Following editors’ recommendation, we included the full name of the ethics committee:

**Method** section (page 7):

“The ethical committee CPP Est-II approved the protocol.” was replaced by “The ethical committee, Comité de protection des personnes Est-II, approved the protocol.”
3) Please include all authors emails on the title page.

All authors’ emails were included in the title section:

**Title** section (page 2):
Authors’ e-mail addresses:

Frédéric Fiteni, fredericfiteni@gmail.com
Marine Jary, jary.marine@yahoo.fr
Franck Monnien, francois.monnien@gmail.com
Thierry Nguyen, tnguyen@chu-besancon.fr
Eric Beohou, eric.beohou@gmail.com
Martin Demarchi, mdemarchi@chu-besancon.fr
Erion Dobi, erion_dr@yahoo.com
Francine Fein, ffein@chu-besancon.fr
Denis Cleau, d.cleau@chi70.fr
Serge Fratté, sfratte@chbm.fr
Virginie Nerich, v1nerich@chu-besancon.fr
Franck Bonnetain, franck.bonnetain@univ-fcomte.fr
Xavier Pivot, xavier.pivot@univ-fcomte.fr
Christophe Borg, christophe.borg@efs.sante.fr
Stefano Kim, stefanokim@gmail.com

Kind regards

Stefano Kim